Author’s response to reviews

Title: Rhabdomyolysis due to the additive effect of statin therapy and hypothyroidism. Manuscript type: Case report Authors: Ekrem Yeter, M.D.; Levent Ozdemir, M.D.; Telat Keles, M.D; Engin Bozkurt, M.D Department of Cardiology, Ataturk Education and Research Hospital, Ankara, TURKEY Running Title: Rhabdomyolysis induced by atorvastatin and hypothyroidism Key Words: Statin, rhabdomyolysis, hypothyroidism Corresponding Author: Ekrem Yeter, M.D. Department of Cardiology Ataturk Education and Research Hospital, 06800, Ankara, Turkey Tel: +90-312-2912525 Fax: +90-312-2912705 E-mail: ekremyeter@hotmail.com

Authors:

Ekrem Yeter (ekremyeter@hotmail.com)
Levent Ozdemir (drlevet@yahoo.com)
Telat Keles (telatkeles@hotmail.com)
Engin Bozkurt (ebozkurt@atauni.edu.tr)

Version: 6 Date: 6 April 2007

Author’s response to reviews:

Dear Editor,
We revised our according to your recommendations and instructions.
Kind regards
Ekrem Yeter, MD